Author's response to reviews

Title: Neurofibromatosis Type 1 associated with papillary thyroid carcinoma incidentally detected by thyroid ultrasonography: a case report

Authors:

Bu Kyung Kim (79kyung@hanmail.net)
Young Sik Choi (yschoi@kosinmed.or.kr)
Sangeon Gwoo (desiredream@hanmail.net)
Yo-Han Park (parkjp@kosinmed.or.kr)
Song I Yang (tonibin@paran.com)
Jeong Hoon Kim (gskjh@hanafos.com)

Version: 2 Date: 31 January 2012

Author's response to reviews: see over
Cover Letter

Dear Chief of Editor

We have recently experienced a special case. The title of our manuscript is “A case of neurofibromatosis Type 1 associated with papillary thyroid carcinoma incidentally detected by thyroid ultrasonography”. We wish to submit for publication in the Journal of Medical Case Reports. It is the first case of PTC detected incidentally by US in an NF1 patient.

In this case, there is an unexpected association between diseases. And this case shows the possible pathogenesis of papillary thyroid carcinoma with neurofibromatosis.

Your kind consideration of this paper would be greatly appreciated

Sincerely yours,
Young Sik, Choi, M.D.
professor